

Received 9 February 2024, accepted 3 March 2024, date of publication 8 April 2024, date of current version 9 May 2024.

Digital Object Identifier 10.1109/ACCESS.2024.3386175

## **RESEARCH ARTICLE**

## Investigating the Potential and Challenges of Learning Analytics Tools in Brazilian Education Through Developer Insights

JONES BARONI FERREIRA DE MENEZES<sup>®1</sup>, EMANUEL MARQUES QUEIROGA<sup>2</sup>, ROBERTO MUÑOZ SOTO<sup>®3</sup>, VIRGINIA RODÉS PARAGARINO<sup>®4</sup>, VINÍCIUS FARIA CULMANT RAMOS<sup>5</sup>, LYDIA MENESES DE CARVALHO<sup>®6</sup>, AUGUSTO DE CARVALHO FERNANDES DE SOUZA<sup>6</sup>, AND CRISTIAN CECHINEL<sup>®6,7</sup> <sup>1</sup>Faculdade de Educação de Crateús, Universidade Estadual do Ceará, Cratéus 63700-000, Brazil

<sup>2</sup>Instituto Federal de Educação, Ciência e Tecnologia Sul-Rio-Grandense (IFSul), Pelotas 96015-560, Brazil

<sup>4</sup>Institute for the Future of Education, Tecnolótico de Monterrey, Monterrey 64700, Mexico

<sup>5</sup>Departamento de Engenharia do Conhecimento (EGC), Universidade Federal de Santa Catarina, Florianópolis 88040-900, Brazil

<sup>6</sup>Centro de Ciências, Tecnologias e Saúde, Universidade Federal de Santa Catarina, Araranguá 88040-900, Brazil

<sup>7</sup>Programa de Pós-Graduação em Computação, Centro de Desenvolvimento Tecnológico, Universidade Federal de Pelotas, Pelotas 96010-610, Brazil

Corresponding author: Roberto Muñoz Soto (roberto.munoz@uv.cl)

This work was supported in part by Brazilian National Council for Scientific and Technological Development—CNPq under Grant 409633/2022-4 and Grant 305731/2021-1, and in part by Project ANID/FONDECYT/REGULAR/1211905.

This work involved human subjects or animals in its research. Approval of all ethical and experimental procedures and protocols was granted by Resolution 510/2016 and Circular Letter No. 2/2021 from the National Commission of Ethics in Research (CONEP).

**ABSTRACT** Learning analytics (LA) tools hold promise for transforming educational practices, providing insights into student performance, and aiding instructional planning. Despite the growing body of literature, the integration of these tools into daily educational processes remains a complex challenge. Latin America has experienced a significant expansion in its literature on Learning Analytics, with Brazil being one of the most important exponents. This study investigates the viewpoints of developers behind learning analytics tools in Brazil, in order to understand their perceptions about the potential and limitations of the tools. Using a multiple case study design with a qualitative approach, we conducted in-depth interviews with six developers actively involved in creating learning analytics solutions. Our findings reveal three overarching categories: (i) Functionality of the Learning Analytics Tools, which includes applications such as predicting dropout risks, automated feedback, and didactic course planning; (ii) Motivations and potentialities, highlighting integration with academic systems and the facilitation of pedagogical interventions; and (iii) Limits and barriers, identifying challenges from socioeconomic, technical, and pedagogical perspectives. This research underscores the significance and potential impact of learning analytics tools in the educational landscape. As a next step, we recommend expanding the scope of investigated tools and further exploring the effective utilization of analyzed data, as well as assessing the overall effectiveness of interventions. This study contributes valuable insights to both the academic discourse on learning analytics and the practical implementation of these tools in educational settings, particularly within the context of Brazilian developers' perspectives.

**INDEX TERMS** Latin America, Brazil, learning analytics adoption, qualitative research.

The associate editor coordinating the review of this manuscript and approving it for publication was Meriel Huggard<sup>10</sup>.

<sup>&</sup>lt;sup>3</sup>Escuela de Ingeniería Informática, Universidad de Valparaíso, Valparaíso 2362735, Chile

#### I. INTRODUCTION

Over the past two decades, we have witnessed significant changes in our communicative and social structures. Scientific and technological advances have given rise to a new form of society, the information society. This new social organization is based on the integration of people through digital networks that employ Digital Information and Communication Technologies (DICT).

This trend can be observed in different domains, including education. Although the use of DICT in the educational process is not new, its widespread adoption has become more evident during the COVID-19 pandemic. The need for social distancing to curb the spread of the virus has required the continuation of pedagogical activities through virtual means, mediated by digital technologies [1], [2].

The digital age has led to the expansion and improvement of computer systems, which includes managing the teaching and learning process. Learning Analytics (LA) has been collaborating with this process, using both static and dynamic data from students and their contexts to support decision-making in learning and teaching [3], [4].

According to the Society for Learning Analytics Research (SoLAR), LA can be defined as the "measurement, collection, analysis, and communication of data about learners and their contexts, for purposes of understanding and optimizing learning and the environments in which it occurs" [5], a definition that is adopted in this investigation.

The benefits of LA include: 1) improving the efficiency of the overall institutional functioning; 2) strengthening the regulation of the teaching and learning environment, leading to positive impacts on practice; and 3) providing teachers with methods and tools to perform their tasks more effectively [6]. Learning Analytics draws on principles from various disciplines, including computing, statistics, social sciences, pedagogy, psychology, and education, as noted by [7] and [8].

Taking into account the above, LA can serve as a vital instrument for modifying learning projects and supporting teachers and managers in (re)planning, with the ultimate goal of improving pedagogical processes and improving learning outcomes. However, the implementation of LA tools is not without challenges, and it remains limited to a small number of institutions [9], [10], [11], [12]. Therefore, understanding the adoption of LA tools in educational institutions and how they help to make administrative and pedagogical decisions is crucial [13], [14].

Latin America has also experienced an expressive grown in this field, with Brazil ranking first in the absolute number of scientific papers related to LA in Latin America and fifth in scientific papers per million inhabitants [15]. This valuable academic contribution emphasizes the importance of adopting LA while being aligned with institutional principles and the learning design of disciplines [12].

However, despite the significant number of publications, the implementation of LA in the daily practice of the educational process is more complex. According to [14], the implementation of ICT in education in Brazil faces several barriers, such as i) lack of government investment, ii) insufficient digital literacy of teachers, iii) scarcity of human resources trained in handling technologies, and iv) disparities between public and private education. [16] pointed out that, even with large amounts of data collected about students' digital traces, most of the LA solutions implemented in Higher Education Institutions in Brazil still are of small scale. Taking into account the scarcity of empirical studies that examine the practical and real possibilities of using LA [17], this investigation intends to better understanding this aspect from the perspective of LA developer in Brazil. The following research questions are proposed:

- **RQ1**: Which are the characteristics of some Brazilian Learning Analytics tools under usage nowadays and how they relate to the literature of the field ?
- **RQ2**: How do developers of Learning Analytics (LA) tools in Brazil view the potential and limitations of these solutions?

Based on this problem statement, this study employs a qualitative approach with a multiple case study design conducted through interviews. The central objective is to understand the developers' perspectives on the application possibilities and limitations of LA tools developed in Brazil. The remainder of this paper is organized as follows. Section II presents the relevant literature on the implementation of LA in practice. Section III outlines the research methodology adopted in this study, and Section IV presents the findings of the analysis. Section V discusses the results in light of existing literature and experiences. Finally, Section VI concludes the study and suggests directions for future research while also highlighting its limitations.

#### **II. RELATED WORKS**

In recent years, there has been a growing body of research on Learning Analytics (LA), as evidenced by the works of [18], [19], [20], and [21]. However, there is a pressing need for further exploration of the practical implementation of LA, particularly in the context of higher education institutions. This necessity arises from the presence of various sociotechnical challenges, as highlighted in the studies of [10], [12], and [22].

Despite the growing interest and research in the field, the impact of LA in the real world is still limited, particularly in Latin American and developing countries, where small-scale adoption predominates [15]. This phenomenon is attributed to the complex nature of educational systems, the emphasis on analytics over learning, high demands on technological infrastructure and resources, stakeholder engagement, and concerns regarding ethics and privacy of the data [23].

In the initial stages of its development, research in the field of Learning Analytics was predominantly concentrated in North America, Europe, and Australia. A pivotal turning point occurred in 2017 when [24] demonstrated that Latin

American universities had initiated the analysis of teaching and learning processes through Learning Analytics, albeit at a relatively small scale. Subsequent works emphasized the significance of comprehending Learning Analytics tools in Latin American countries, highlighting the region as promising for the field's advancement [25], [26].

Over recent years, various scientific initiatives have played a crucial role in fostering the growth of Learning Analytics in Latin America. Notable examples include the LALA project (Building Capacity to Use Learning Analytics to Improve Higher Education in Latin America) [27], the LALA conferences (Learning Analytics for Latin America) [28], and dedicated special editions in prestigious journals such as the British Journal of Educational Technology [29] and the Journal of Learning Analytics.<sup>1</sup> These endeavors collectively contribute to the heightened visibility and advancement of Learning Analytics within the Latin American context. As indicated by [15], Brazil holds a significant position in the region, leading in the number of published papers among the countries. Additionally, Brazil maintains robust collaborative connections, extending its partnerships not only within Latin America but also with countries on other continents.

In recent years, there has been notable growth in research focused on the adoption of LA, particularly within higher education institutions. The majority of these studies have mostly concentrated on the contexts of Europe, North America, and Australia (e.g. [30], [31], [32]). As an illustration, [33] conducted a review of 18 distinct Learning Analytics (LA) adoption frameworks, pinpointing five primary challenges encountered by teachers when incorporating LA in classroom settings. These challenges include: i) difficulties in integrating technical and pedagogical expertise in LA use; ii) a lack of connection between LA and educational theories or pedagogies; iii) the failure to align LA with teachers' practice; iv) ethical and privacy concerns; and v) the burden of additional workload and a shortage of time.

In Latin America, a discernible shift has occurred, characterized by an increasing number of studies exploring the adoption of LA in educational institutions. For instance, [12] investigated perceptions and concerns surrounding the adoption of Learning Analytics (LA) in four Latin American universities. The findings highlighted key needs associated with LA implementation, encompassing: i) a demand for quality feedback and data-driven support from teaching staff to enhance learning outcomes; ii) the necessity for data to inform support interventions; iii) a call for timely alerts from managers to better support struggling students; iv) the importance of meaningful performance evaluation of teaching quality; and v) the requirement for actionable information from staff to assess the effectiveness of support interventions. In the same direction, [34] delineates 13 significant themes addressing current obstacles in the practical application of Learning Analytics (LA) tools. The author proposes categorizing these topics into five main areas: i) data

<sup>1</sup>https://learning-analytics.info/index.php/JLA/announcement/view/181

management; ii) administration and training; iii) pedagogical support; iv) data analysis; and v) legislation, privacy, and law. This data elucidates more concrete actions that can assist institutions in adopting LA, particularly by pinpointing the necessary institutional measures, the individuals involved, and their respective responsibilities.

In the Brazilian context, [13] investigated the expectations about LA from a Brazilian Higher Education Institution and considering three different stakeholder groups: students, teachers, and teachers assistant. Here, even though all participants agreed on the potential of LA to improve learning and personalized processes, they mentioned the need for simple and powerful visualizations, and suggestions of pedagogical actions by the LA tool as requirements for the adoption of LA solutions, and the risk of invading students privacy as one of their main concerns. Continuing within the context of Brazil, an initial survey carried out by [35] in the south and southeast regions with Higher Education Institutions (HEIs) revealed that only half of the participant institutions have implemented some form of Learning Analytics (LA) solution. Moreover, merely 6% of them employed these solutions systematically.

Given the prevalent challenges associated with Learning Analytics (LA) adoption in Latin America, coupled with its nascent implementation in Brazilian institutions, this paper aims to complement existing literature by delving into the challenges and possibilities of LA solutions in Brazil. The unique perspective offered here stems from the insights of the creators and developers behind these tools. As observed previously, much of the existing literature on LA adoption primarily focuses on the perceptions of students and teachers. To our knowledge, this work represents the first attempt to compile insights directly from the creators and developers of these solutions.

#### **III. METHODS**

The research paradigm encompasses the theoretical and practical frameworks that guide the constitution and execution of the investigation, leading to the production of new scientific knowledge. The choice of the research paradigm is not only determined by the research problem but also by the researcher's assumed intersubjectivities [36].

Based on the objectives and the chosen paradigm, this investigation is based on a qualitative approach. Qualitative research is ideal for studying subjective aspects of a certain group of participants, enabling a deeper understanding of a specific topic and allowing researchers to comprehend the complexity of the studied phenomenon [37].

Referring to the research methods, a qualitative approach using a multiple case study design was employed, and data were collected through semi-structured virtual interviews. The interview guide consisted of 17 open-ended questions (available at https://bit.ly/LAtoolsinterview) that aimed to explore the identification, functionality, data collection, analysis, and intervention of LA tools. The questions also addressed the participants' motivations, perceptions,

Identification 01 Respondent's name Country State Type of organization (Ø) STEP Learning analytics tool 02 Name Usage time Target Audience Number of users served ethical care Training for using the tool ناته **Tool Operation** 03 Data sources and types Participants in the data analysis process Institutional policy STEP Potentialities Π4 Tool capability level Means of intervention • • • STEP 05 Limitations and barriers Main limitations and barriers in the use of the tools

FIGURE 1. Steps in the interview script.

limitations, and barriers regarding the use of LA tools (see Figure 1).

The semi-structured virtual interviews were conducted remotely via the Google Meet web conference platform, and were recorded with the participants' consent during the months of January and February 2023. Once the recordings were collected, they were transcribed, and qualitative data was analyzed using the content analysis method proposed by [38]. Content analysis plays a fundamental role in scientific research, providing a robust methodology for exploring, interpreting, and understanding the content of different forms of human expression. Its application is broad, and its flexibility makes it a valuable tool for researchers seeking rich and contextualized insights [39], [40], [41]. It is widely used in educational research and can even be employed in mixed-methods research [42].

This method involves three stages: i) data preparation; ii) thematic coding and categorization; and iii) descriptive and/or comparative analysis. To facilitate qualitative analysis,

Data analysis tools/software

Motivation(s) for using the tool in educational management

statistical software tools, namely NVivo® (version 11) and Iramuteq® (version 0.7 alpha 2), were used. A sentiment analysis was conducted as the fourth stage to gauge the researchers' sentiments concerning their tools. Figure 2 illustrates the comprehensive methodology used in the study.

The research sample was composed of six Brazilian researchers who were actively engaged in Learning Analytics projects and had experience in research and development in Higher Education Institutions. The identification of participants was based on a review of literature on Learning Analytics in Brazil and the Workshop on Practical Applications of Learning Analytics in Educational Institutions in Brazil [43]. The six participants (refer to Table 1) were chosen through a purposive sampling technique and assigned codes (D1, D2, D3..., D6). The sample comprised five men and one woman, all of whom volunteered in response to an invitation. These participants possessed substantial expertise in developing and implementing computational solutions for Learning Analytics, along with the ability to train and educate others in this domain.

To ensure compliance with ethical recommendations for scientific research, this investigation adhered to the standards recommended in Resolution n° 510/2016, which is specific legislation for investigations in the areas of Human and Social Sciences [44], and Circular Letter No. 2/2021 of the National Research Ethics Commission in Brazil, which provides guidelines for research in virtual environments. At the beginning of the recorded interviews, participants provided consent by reading and agreeing to the points contained in the Free and Informed Consent Form.

The ethical principles assumed in this research are also in accordance with the American Educational Research Association (2011) and with what is advocated by [45], which include professional capacity, integrity, social, professional, educational, and scientific responsibility, respect for the rights, dignity, and diversity of participants. Confidentiality, transparency in storage and handling, as well as precision in the use of data, are prioritized, without falsifying data, fabricating results, or plagiarism. All the participants also agreed this paper directly cited their respective LA tools and papers.

#### **IV. RESULTS**

After transcribing the interviews, a textual statistical analysis was conducted using the qualitative data analysis software Iramuteq $(\mathbb{R})$  (version 0.7 alpha 2). The use of Iramuteq software assists in the treatment of textual data and offers various analysis possibilities based on text statistics or lexicometry. This process involved lemmatization, which searches and relates words by their root, ignoring tense, gender, plural, and supplementary formations [46].

The general corpus was formed from six transcripts, totaling 682 text segments aligned for analysis. The examination identified a total of 21,511 occurrences of words and forms, with 2,084 being distinct words and 959 occurring only once. In Figure 3, the frequency distribution of words used



**FIGURE 2.** Stages followed in the methodology.

TABLE 1.	Profile of the	researchers	participatin	in the	research.

Respondent	Education	Type of Linked	Brazilian Region where	
Code	Education	Institution	the institution is located	
D1	PhD in Electrical Engineering. Master in Electrical	Federal Public	North East	
DI	Engineering. Degree in Computer Engineering	rederar r done		
D2	PhD in Computer Science. Master in Computer	Private non profit	South	
D2	Science. Bachelor in Informatics.	Thvate non-pront		
D3	PhD in Computer Science. Master in Informatics	Federal Public	North	
05	and Graduation in Data Processing.	rederar r done		
D4	PhD in Computer Science. Master in Computer	Federal Public	North East	
D4	Science. Degree in Computer Science.	rederar r done		
D5	PhD in Computer Science. Master in Computer	Federal Public	South	
05	Science. Bachelor of Systems Analysis	rederar r done		
D6	PhD in Computer Science. Master in Computer	Federal Public	North East	
	Science. Bachelor's Degree in Computer Science.	reactar rublic		

in the interviews is depicted. The coordinates represent the logarithmic function of the order of occurrence and the logarithmic function of the frequency of these words.

The figure illustrates a pattern where numerous words are repeated a few times, while only a limited set of words are repeated frequently.

A word cloud was generated (Figure 4) to identify the most frequently used words in the interviews. The top five words in terms of frequency were "Student" (Estudante), "Teacher" (Professor), "Tool" (Ferramenta), "Data" (Dado), and "School" (Escola).

Through the utilization of a similarity analysis (Figure 5), it became apparent that these words exhibit a robust interconnection, consistently appearing together in the word cloud. The analysis unveils occurrences and relationships between words, contributing to a more comprehension of the textual corpus. Among the statements made by the interviewees, "Student," "Teacher," "Tool," and "Data" emerge as the most prominent terms. Elaborating further can yield additional words and enhance the articulation of their relationships.



FIGURE 3. Analysis of the number of repeated words in the textual corpus.

The terms "develop" (desenvolver) and "use" (usar) are closely tied to the tool, representing integral processes in



FIGURE 4. Word cloud of the most frequently occurring words in the interview transcripts.



FIGURE 5. Relationship between most frequent words in the interview transcripts.

the practical application of learning analytics tools. These tools played a pivotal role in providing data for diagnosing "problems" (problema) and constructing predictive dropout and failure "models" (modelo). Moreover, these tools proved advantageous for educators in both "universities" (universidades) and "schools" (escolas), facilitating a deeper understanding of students' learning trajectories. Through the analysis of student activities and behaviors, teachers can intervene more effectively and with greater precision.

The integration of these four terms is vital in the learning analytics (LA) processes, as the primary goal of LA is to enhance student learning. By employing LA tools to analyze educational data, teachers can extract valuable insights. This, in turn, empowers them to make informed decisions, facilitating targeted interventions that support and improve student learning.

Two axes of analysis are proposed based on the research objectives and the words that emerged from the analysis.

- 1) Characteristics of the Learning Analytics Tool this axis encompasses the characteristics related to the target audience, means of data collection and analysis, ways of presenting the analyzed data, and possible means of intervention:
- 2) Potentialities and Limitations focuses on the motivations for developing and adopting the tool, as well as its potential for teachers and students in educational and learning management, in addition to the main obstacles to the adoption of the Learning Analytics tool. These axes align with the literature in the field and can aid in further exploration of LA in education.

A total of 124 codes were identified, comprising 82 codes related to the characteristics of the learning analytics tool (Axis I) and 42 codes associated with the potentialities and limitations of the tool (Axis II).

Hence, this section is divided into two subsections, each corresponding to the axes. Initially, we will delve into the presentation of Learning Analytics (LA) tools, with a focus on identifying their central objectives, target audience, applicable teaching modalities, and their current usage status. Additionally, we will underscore the functioning of data collection, analysis, and intervention mechanisms employed by LA tools. Following that, we will scrutinize the motivations, perceptions, limits, and barriers surrounding the use of these tools. Ultimately, we will conduct a comprehensive textual analysis, encompassing the transcription of the interviews and the identification of key words present in the dialogues. Additionally, we will explore the relationships between the interviewees.

#### A. CHARACTERISTICS OF THE LEARNING ANALYTICS TOOLS

In this multiple case study, we identified six learning analytics tools (LAT) developed by Brazilian research groups (Table 2). These computational solutions are applicable to both face-toface and distance education, and are designed for students, teachers, and administrators. The main objective of these tools is to improve learning outcomes through features such as automatic grading, feedback provision, prediction of dropout and failure risks, and didactic planning of disciplines/courses.

The main motivations for developing these solutions include personal perspective and/or development opportunities, improvement and advancement of the educational process, facilitation of teaching work, promotion of learning, and educational management. These categories are supported by the statements of the developers, which are listed below:

- {...} identify the student who has a high level of risk of dropout and low performance throughout the course, *before it becomes a problem for the student* {...} (D2)
- {...} bring a tool that interactively throughout the semester would indicate clues to teachers that could then be worked on properly  $\{\ldots\}$ , which would meet teachers' needs {...} (D3).

TABLE 2. Details of the learning analytics tools developed in Brazil.

Tool	Central	Target	Teaching	Status	
Code	objective	Audience modality		Status	
LAT1	Risk of Aband. and Dropout	Teachers	Face-to-face	In usage	
LAT2	Risk of Aband. and Dropout	Teachers	Face-to-face and distance education	Discont.	
LAT3	Activity cor. and feedback	Students	Face-to-face	Discont.	
LAT4	Inst. design planning	Teachers and Manager	Face-to-face and distance education	In tests	
LAT5	Risk of Aband. and Dropout	Teachers and Manager	Face-to-face and distance education	In usage	
LAT6	Cor. of textual activity	Teachers	Face-to-face	In usage	

- {...} I would like to work with formalism {...}. So I started researching what I could work on in education. And Learning Analytics came to me. I was enchanted and I said that's what we need because we have a lot of data and we don't use anything {...} (D4)
- {...} the quality of education improves, we want to train better students for this, we need to have teachers who better understand what is happening in the classroom so that we can improve within the contents {...} (D5)

To offer a more comprehensive insight into the learning analytics tools, Table 3 delineates the specific methodologies employed by each tool in data collection, analysis, and intervention. The table underscores commonalities in the approaches utilized for collecting, analyzing, and presenting data, thereby facilitating the intervention processes.

Furthermore, Table 4 illustrates the practical applications of the learning analytics tools developed in Brazil across diverse teaching modalities (including face-to-face and distance learning) and various educational levels (ranging from primary and secondary to higher education).

Considering the information outlined in Tables 3 and 4 pertaining to the content of each tool, a cluster analysis was executed to discern similarities in their central objectives and operational characteristics. The primary objective of this analysis was to highlight the proximity among the tools. To accomplish this, the Pearson correlation coefficient was computed to gauge the correlation between the tools. The resultant correlation coefficients were then utilized to create clusters, where elements within the same group

TABLE 3.	Specifics of	of data coll	ected and	analyzed b	oy learning	analytics
tools orig	ginating from	m Brazil.				

Tool	Data	Data made available	
Code	source	post-analysis	
	Public	Dropout risk percentage	
LAT1	detebese	per class and per	
	uatabase	student	
	Virtual loarning	Dashboard with a	
LAT2	virtual learning	graphical view of a	
	environment	series of indicators	
Virtual learning		Quantitative and	
LAIS	environment	qualitative data.	
LAT4	Virtual learning	Quantitativa data	
LAI4	environment	Quantitative data	
		Dashboard with a diversified	
I ATT5	Virtual learning	graphical view (temperature,	
LAIJ	environment	dispersion, line and column	
		graphs)	
LAT6	Public database	Quantitative data	



**FIGURE 6.** Hierarchical tree of similarity between the description of learning analytics tools.

exhibit similarity, while those in different groups manifest dissimilarity. The correlation coefficient data is presented in Table 5, and the hierarchical clustering is depicted in Figure 6.

Upon analyzing the hierarchy, it becomes evident that the tools are primarily distributed based on their applicability. LAT1, LAT6, and LAT5 are either inactive or in initial testing stages, while LAT2, LAT3, and LAT4 are in practical use.

LAT1 is predominantly used in primary education to evaluate the risk of student dropout and inform both the management group and teachers about each student's status. It was developed as a component of the National Policy for the Recovery of Learning in Basic Education, established to address the deficits resulting from the pandemic in the years 2020 and 2021. The data is acquired through a psychometric instrument, and the information is stored in a database managed by a public institution in Brazil. The Ministry of Education has access to this database and employs learning analytics (LA) tools to analyze the data across various dimensions of the questionnaire. This analysis aids in identifying potential dropout risk factors. Schools receive

#### TABLE 4. Characteristics of learning analytics tools developed in Brazil.

Tool Code	Description	References
LAT1	An Early Warning System (SAP) to identify students at risk of dropping out of school. This system utilizes an Assessment Instrument for Relational Risk Factors for School Dropout (IAFREE, in Portuguese), which identifies five key relational factors for predicting school dropout risk: I. Student-School, II. Student-School Professionals, III. Student-Family, IV. Student-Community, and V. Student–Student. The study involved a sample of 15,924 Brazilian high school and middle school ; students. The findings underscore the effectiveness of the IAFREE model as an early warning system, providing a more precise and comprehensive understanding of the elements contributing to school dropout.	[47]
LAT2	Identified the potential risk of school dropout for students from 19 classes, from 4 distance learning courses. The teachers and tutors involved with the case study had access to the proposed system with dropout prediction and based on this information, triggered specific welcoming and attention actions for students. The data shows that after using the application there was a reduction in dropout rates in subjects (average of 0.33 dropped to 0.27) and an increase in the approval rate (0.57 to 0.61). In both cases with statistical significance.	[48] [49]
LAT3	It is an online judge with the purpose of automating the correction of programming exercises. Teachers can provide programming exercises to their students, who in turn can code solutions and submit them through the system interface. As soon as the student submits a solution for a given exercise, the tool informs whether it is correct or not. If the system notifies that the solution code is not correct, whether due to syntax or logic errors, the student will be able to review their code and resubmit it as many times as desired, up to the deadline stipulated by the teacher. This procedure provided an increase in approval rates, which jumped from 50% to 70% among the experimental classes, while it remained at around 30% in the control classes.	[50]
LAT4	It is a free and open-source tool allowing integration with VLEs such as Moodle. It allows one to build the Instructional Design of a course, with its sequencing, parallel or competing activities and also restrictions on the availability of such activities. Furthermore, it is possible to monitor, in real time, student performance in each activity, specifically related to quiz analysis, module access and activity history, and adapt the course to student needs. The tool is in the development phase and the next steps are its validation with real courses.	[51] [52]
LAT5	A cloud-based learning analytics service designed for Moodle enables teachers and course managers within the Virtual Learning Environment (VLE) to access student risk predictions and visualizations of their interactions in the course. This functionality is achieved through a Moodle-associated plugin (block) that gathers interactions from system logs and generates predictive models. These models are then presented on the Dashboard through graphical visualizations, facilitating the interpretation of results.	[53]
LAT6	The tool analyzes students' written production to support teachers in providing more effective and pedagogical feedback in the learning process. The analyzed elements include coherence, cohesion, readability, and the content of the essay, using Natural Language Processing (NLP) techniques.	[54] [55]

reports that showcase the results of this analysis, enabling them to intervene more effectively and mitigate the risk of student dropouts.

The tools **LAT2** and **LAT5** share a common objective, which is to assist higher education students in both face-to-face and distance learning settings. However, it should be

noted that these tools have only been tested and evaluated in the context of distance education. **LAT2** is a tool developed in collaboration with an artificial intelligence-specialized startup and the graduate programs of an educational institution. This tool is integrated into the Moodle virtual learning environment, capturing data from digital traces of distance

	LA1	LA2	LA3	LA4	LA5	LA6
LA1	1	0,64	0,66	0,66	0,88	0,95
LA2		1	0,91	0,86	0,67	0,63
LA3			1	0,92	0,67	0,65
LA4				1	0,69	0,66
LA5	-	-			1	0,89
LA6						1

 TABLE 5. Pearson's correlation coefficient between the similarities of the words related to the tools.

education students, utilizing approximately 36 variables. Subsequently, the collected data is employed to establish a machine learning model capable of accurately predicting the risk of student dropout. Through the analysis of this data, course coordinators and professors can access information specific to their classes and undergraduate students via a dashboard. This allows them to pinpoint students in need of extra attention on a weekly or daily basis, empowering them to intervene with pedagogical strategies and proactively prevent the perpetuation of dropout patterns or learning difficulties. LAT5 is a tool developed through inter-institutional collaboration and co-creation with a startup. Its main goal is to support teachers in the teaching and learning process within the Moodle Virtual Learning Environment, utilizing a plugin to gather student access and interaction data. Through the analysis of various variables and dimensions within this data, a predictive model is generated, accurately identifying the risk of student failure and dropout.

A pivotal aspect of the virtual educational process involves course planning and instructional design. **LAT4** is a tool specifically crafted to enhance course design and foster effective learning outcomes. Its primary goal is to facilitate graphic instructional design planning for courses and disciplines offered in Moodle, relying on the High-Level Activities Network (Petri Net). This tool serves not only to create new courses but also to import existing Moodle courses, incorporating all registered activities into the tool. This feature empowers instructors to refine graphic instructional design through learning analytics, ensuring that the course design optimally enhances the learning experience.

The monitoring of student performance is streamlined through the use of automated tools that gather real-time data on student access and interaction in Moodle, encompassing quiz results, module access, and activity history. Instructors receive this data through descriptive statistics, enabling them to identify students at higher risk of failure. Armed with this information, teachers can intervene more proactively by conducting targeted outreach, adjusting activities, or even refining instructional design to better meet the needs of their students.

LAT4 is closely linked with two other tools, LAT3 and LAT6, which concentrate on automating activity corrections and delivering automatic feedback. Despite sharing this common objective, the tools diverge in their target audience,

with LAT3 designed for students and LAT6 tailored for teachers.

LAT3 was expressly designed to support teaching and learning in disciplines requiring the creation of computer programming codes. It provides an environment where students can test their code using their inputs and execute it. Upon deeming their code correct, they submit it for verification, and it undergoes correction using a test case as a model. Furthermore, LAT3 offers feedback on any errors or successes in the code. This approach streamlines the efforts of both students and teachers, fostering a more efficient and effective learning process. In contrast, LAT6 is a tool specifically designed for Android phones, employing artificial intelligence to correct textual activities. It is predominantly utilized in face-to-face basic education, with a specific focus on elementary school students, while also serving as a support tool for teachers. By automating the correction process of written assignments, LAT6 streamlines tasks for both students and teachers, allowing them to redirect their time and effort towards other facets of the learning process. In this scenario, the student writes the textual activity on paper, and the teacher captures an image to send it to the tool. The tool then processes the information and provides feedback based on predefined criteria, such as textual cohesion, coherence, and adherence to the topic. The feedback can be tailored to each student individually or compiled for the entire class. Informed by this feedback, the teacher can make decisions that contribute to the ongoing learning of the students.

It is important to note that all tools have ethical precautions associated with the use of data, in compliance with international legislation such as the General Data Protection Regulation (GDPR) in the European Union and the General Data Protection Law (Law n° 13.709/2018) enacted in Brazil in 2018. Data is anonymized, encoded, and encrypted to ensure privacy and security. Furthermore, teachers have exclusive access to their class data, and there is no sharing of data with all users of the tool. The ethical concern associated with the use of learning analytics is increasingly being discussed and implemented in academic research and in the technological and scientific products that result from it. In a recent study, [56] emphasized this concern and demonstrated the need for a relationship between ethics and learning analytics. The authors argue that it is necessary to expand ethical discussions and relate them to the characteristics of the different approaches used in learning analytics. This is important to raise awareness about the role of the researcher and promote conscious and responsible use of data throughout the process of collection, analysis, retention, administration, and intervention. As mentioned above, these learning analytics tools are still in the process of development and/or applied on a small-scale and experimental basis. Although developers are aware of the importance of LATs and show interest in them, there is a need to conduct further research noe research is needed on these tools to analyze and optimize their activities to improve learning and the environments in



FIGURE 7. Potentialities of learning analytics tools.

which where they are used. It is crucial to ensure that the tools are reliable, valid, and effective in their intended use, as well as to assess their impact on student learning outcomes. Therefore, it is necessary to continue to investigate and refine investigating and refining these tools to fully realize their potential in promoting to promote for student success in education. In summary, the examined learning analytics tools examined are versatile, applicable to both face-to-face and distance learning modalities, with a central primary focus on the benefiting of students, teachers, and administrators. These tools primarily excel in automatic and personalized activity correction, providing feedback, predictive modeling of dropout risk, and facilitating instructional design planning for courses and disciplines. To gain a deeper understanding of these computational solutions, the following section will delve into the motivations, potentialities, limitations, and barriers associated with the adoption of learning analytics tools in the daily practices of educational institutions.

### B. POTENTIALITIES AND LIMITATIONS OF ADOPTING LEARNING ANALYTICS TOOLS

Upon scrutinizing the data, we have identified the primary strengths and limitations of learning analytics tools as perceived by their developers. Notably, one potential advantage is that these tools align with the requirements of two pivotal stakeholders in the learning analytics process: educational management and teachers. A more comprehensive exploration of these positive aspects of the tools can be found in the participants' statements and detailed in Figure 7.

- {...} a complete tool, because it was integrated online with the systems that generated the data. Both the academic system and the virtual learning environment system. {...} thus, it has the potential of an intervention process that it can really **intervene and change reality** to improve this reality of both dropout and low performance. (D2)
- {...} the perception I have is always very good, people really like it because it takes a lot of work off the teacher's shoulders {...} (D3).
- {...} I can follow my students graphically and with a dashboard with an accuracy of 90% so that I can know if this student is prone to dropping out {...} (D5)
- {...} a tool that can be used in everyday life, helping with text corrections and giving feedback to the student {...} (D6)



FIGURE 8. Challenges in adopting learning analytics tools in Brazil.

Nevertheless, these tools exhibit limitations, as illustrated in Figure 8. These constraints encompass technological challenges, including the requisite for reliable and efficient platforms, along with human resource limitations, emphasizing the necessity for specialized professionals to operate the tools effectively. Other drawbacks comprise infrastructural deficits, logistical challenges, and the imperative need for incentives to stimulate technological innovation. Moreover, some participants underscored that the non-commitment of education management and a lack of awareness among stakeholders in the educational process can impede the widespread adoption of these tools:

- {...} provide information to students through a *dashboard* {...} (D1)
- {...} access barriers both to the academic system of the university, and access to the Moodle system, for the integration of these systems from a resource that worked effectively, quickly, safely [...] (D2).
- {...} used only for the Computing and programming areas, for the time being {...} (D3)
- {...} awareness of the community in general that learning analytics is within artificial intelligence, but it has the purpose of bringing benefits of giving positive support to decision-making and not as a mechanism that can be used to generate criticism and unrelated discussions productive {...} (D5)

According to the developers, it is imperative for school managers, teachers, and government officials to cultivate awareness and promote the dissemination of information among active participants in the educational process who contribute to the collection of educational data, aiming to enhance education. Furthermore, there is a need for incentives to expedite research on technological innovation, facilitating the training of qualified and skilled human resources for the development and improvement of tools like those mentioned here. Although the results already demonstrate the effectiveness of such tools in improving education, there remains a necessity for further development and widespread dissemination of these technologies to ensure equal and effective access to education for all. Next section will delve into the strengths and limitations of the LA solutions outlined in this study in light of developers' opinions.

Respondent	Neutral	Negative	Positive
D1	0.4727	0.2947	0.2325
D2	0.5009	0.2940	0.2052
D3	0.5481	0.2776	0.1742
D4	0.5167	0.2824	0.2009
D5	0.4844	0.3029	0.2128
D6	0.4876	0.3012	0.2112

#### TABLE 6. Sentiment evaluation by respondents.

#### C. SENTIMENT ANALYSIS

Sentiment analysis was performed over the respondents answers using multilingual XLM-roBERTa-base model [57]. RoBERTa, short for Robustly Optimized BERT Pre-training, is a pre-trained language model developed by Facebook AI, with a specific emphasis on sentiment analysis tasks. Serving as an extension of the BERT (Bidirectional Encoder Representations from Transformers) model, RoBERTa uses a transformer architecture and integrates various modifications aimed at enhancing its overall performance. For each response, the analysis returned the sentiment score for three distinct sentiments: positive, negative and neutral. The scores vary from -1 to 1 where as close they are to 1 the stronger is that given sentiment. The results of this analysis are shown in Table 6.

As illustrated in Table 6, consistent patterns emerge from the sentiment analysis. Upon reviewing the findings, it becomes apparent that all respondents conveyed neutral sentiments during their interviews. Furthermore, when contrasting negative and positive sentiments, it becomes evident that negative sentiments consistently slightly surpass the positive ones, suggesting a slight leaning towards an overall neutral sentiment with a negative undertone. These specific data points and comparisons underscore the uniformity of sentiments expressed among the respondents and may offer insights into the prevailing perspective of LA developers. This phenomenon could be attributed to several factors, such as limited widespread adoption of certain tools or the failure of LA initiatives to achieve the anticipated impact initially envisioned.

#### **V. DISCUSSION**

The main objective of this study was to investigate the viewpoints of developers on the implementation of learning analytics tools in Brazil, with a particular focus on their functionalities, potential benefits, and potential drawbacks. This section is focused on answsering the research questions proposed in the beginning of the paper.

**RQ1:** Which are the characteristics of some Brazilian Learning Analytics tools under usage nowadays and how they relate to the literature of the field?

Six learning analytics tools developed by Brazilian research groups were examined in this multiple case study. These computational solutions cater to both face-to-face and distance education, targeting students, teachers, and administrators. The tools, scrutinized in this study, concentrate on enhancing learning outcomes through features such as automatic activity correction, risk analysis of failure, abandonment, and dropout, as well as didactic planning of courses/disciplines. The functionalities of the analyzed tools align with three out of six scopes advocated by [58], namely prediction (LAT1, LAT2, LAT5), feedback (LAT3 and LAT6), and pedagogy (LAT4). Notably, the remaining three categories—assessment, curriculum, and social learning analytics—fall outside the purview of the investigated tools. This observation opens up new prospects for future local developments in the country, signaling potential areas for expansion and innovation in the field of learning analytics.

Failure, abandonment, and dropout represent significant challenges for the educational system, not only in Brazil but also on a global scale. In Brazil, these challenges were exacerbated during the pandemic period. Data from the National Institute of Educational Studies and Research Anísio Teixeira (INEP), responsible for the Census of Basic Education and Higher Education, reveals a troubling trend. Dropout rates in Secondary and Higher Education doubled compared to the rates observed in 2020 and 2021 [44]. In this context, it is imperative to investigate solutions that can proactively prevent or indicate the likelihood of student dropout, allowing for early intervention to avert the issue. Predictive analysis, leveraging student data from learning management systems or educational process management systems, plays a pivotal role in this regard. This approach is instrumental in optimizing the management process and fostering the dissemination and assimilation of knowledge. By identifying potential challenges early on, educational institutions can implement targeted strategies to support students and enhance overall academic success. The learning analytics (LA) tools examined in this study are aligned with the challenges outlined and draw upon solutions proposed by other researchers, such as [48], [59], [60], [61], [62], and [63]. These tools also illustrate instances of predictive models characterized by a high level of accuracy in anticipating the risk of student dropout. Encompassing both face-to-face and distance learning, as well as basic and higher education, these studies offer valuable insights through dashboards. These dashboards enable students, teachers, and management groups to actively monitor learning progress and assess the potential risk of failure and dropout.

Furthermore, findings from our study underscore the significance of aligning didactic-pedagogical planning of courses with learning analytics. This alignment encompasses various stages, including course design, evaluation, activity correction, and feedback—areas directly addressed by three of the learning analytics (LA) tools examined in this study. As asserted by [37], the relationship between learning analytics and course design should be reciprocal, fostering collaborative reflection on teaching practices and facilitating the identification of strategies that enhance the learning experience. Moreover, providing purposeful feedback on student activities is critical for their academic performance,

and automated and personalized feedback can make this process more accurate and effective. Learning analytics tools have emerged to provide feedback to the most appropriate students through learning indicators, artificial intelligence [64], [65] and dashboards [66]. Moreover, the LA tools examined in this study effectively address the diverse needs of Latin America. For instance, as reported in [12], Latin American students express a desire for quality feedback and faculty support to enhance their learning outcomes. Teachers underscore the importance of timely alerts about students' difficulties, not only to provide effective support but also to make informed pedagogical decisions and evaluate their own practices based on concrete evidence. These needs are also corroborated by the findings of [13]. Additionally, [12] notes that the management team highlights the lack of quality and actionable information for interventions, as well as the imperative to analyze the effectiveness of these interventions when implemented.Notably, these are all characteristics addressed by some of the LA tools scrutinized in this analysis.

# **RQ2:** How do developers of Learning Analytics (LA) tools in Brazil view the potential and limitations of these solutions ?

The potentialities identified in the study converge on the improvement and advancement of academic management and teaching practices. For academic management, LA tools have the potential to be integrated into academic systems and virtual learning environments, enabling better student monitoring, sometimes in real-time, with the aid of dashboards for easier visualization of data analysis. This can lead to the creation of more effective intervention strategies. As for teachers, potentialities lie in the facilitation and improvement of teaching activities, such as better planning of disciplines/courses, automated correction of activities, and personalized feedback to students.

Despite the potential of learning analytics tools, the developers mentioned limitations and barriers that hinder their widespread adoption, especially due to various sociotechnical and pedagogical challenges [12], [22], [67]. The social-technological perspective highlights the issue of social inequality and lack of digital access in Brazil, as well as the budget cuts for research and scientific development. Brazil is a continental country with different regions that present distinct and unequal sociodemographic characteristics. These factors contribute to the challenge of implementing learning analytics tools in a way that benefits all students equally.

According to [44], the Continuous National Household Sample Survey shows a significant disparity in internet access, especially in rural areas, where only 55.6% of the population has access to the internet. The situation is even more concerning in the North (38.4%) and Northeast (51.9%) regions, which also have the lowest per capita household income, approximately 11.2% lower than the national average. In contrast, the South, Southeast, and Midwest regions have the highest average per capita household income. These regional differences in income and access to technology highlight the socioeconomic disparities that exist in Brazil.

The developers of the LAT1 and LAT6 tools, which use basic education data for their analyses and decisionmaking, have identified several barriers to the widespread use of their tools. One major impediment is the lack of cell phones to digitize and send textual productions for analysis. Additionally, the absence of internet connectivity in schools interferes with the sending of socio-demographic data and academic performance of students to the platform, which serves as a crucial source of data for learning analytics.

In order to achieve more accurate performance analysis, it is crucial to consider the socioeconomic profile, conditions, and geographic context of students within the educational process. This is necessary to define the proposed activities, forms of evaluation, and communication, as social inequalities can affect students' basic skills and their access to necessary infrastructure [13].

It is important to note that there is a lack of support for Brazilian technological innovation research, which hinders national technical-scientific development and progress. In recent years, science has been subjected to significant budget cuts, which have directly impacted research, grants, and related activities.

According to data from the Brazilian Society for the Progress of Science [68], there has been a reduction of around 37.6% in the amount allocated to scientific research in the Federal Government's Annual Budget Law from 2019 to 2022. This reduction has had a direct impact on research, grants, and related activities. The reduction in support for research on technological innovation in Brazil carries substantial consequences for the nation's progress, leading the developers of LAT2 and LAT4 to halt their efforts in enhancing their tool.

Infrastructure, time, ease of use, usefulness, personal perspectives, experiences, behaviors, skills, and abilities have been identified as factors that limit the adoption of new technologies [69] and directly impact the technical limitations imposed on the adoption process. Our study also identified technical limitations, such as the restriction of a specific tool to an area of knowledge (Computing and Programming) and students not having access to the results of data analysis, which prevents them from monitoring their own academic performance and learning. This concern is relevant, as many learning analytics tools prioritize presenting their results to teachers and management, without involving students in the implementation process. However, the central focus of learning analytics is to support students and enhance learning outcomes [10].

Providing students with access to a graphical and intuitive dashboard that presents clear and precise information on the various variables used in the analyses is essential for feedback and student ownership of their learning path and performance. Institutions must provide training to enable students to use data as a support for learning [70]. This could also encourage new innovative initiatives for LA tools, which were identified as a limitation of LA adoption for some tools in this study. It is necessary to promote and expand research to sensitize the population about the importance and potential of LA. This concern is not unique to Brazil, as evidenced by [10] who carried a study on the adoption of LA in 83 higher education institutions across 24 European countries. The study highlights that the biggest challenges associated with the adoption of LA are related to tensions and concerns around exploring innovative ways to combine skills, funding, infrastructure, people, and the culture of data use while maintaining academic efficiency and financial management.

Many of the highlighted limitations in our study align with various categories influencing the adoption of Learning Analytics as emphasized by [71]. The authors identified 14 factors distributed across 6 major groups that influence the adoption of Active Learning (AL) in higher education: User involvement and training (UIT); Project management and control (PMC); Skill mix (SKM); Organization fit (ORG); Technology planning (TEC); and Software system design (SSD). Our study identified limitations correspond to several factors outlined by the authors, encompassing all interconnected categories. Specifically: i) challenges related to technological aspects correspond to the PMC, TEC, and SSD categories; ii) issues such as the lack of infrastructure, logistics, and incentives for technological innovation are acknowledged in PMC, ORG, TEC, and SSD; iii) problems related to qualified human resources are associated with UIT, SKM, and ORG; and iv) the absence of commitment to managing education and raising awareness among educational stakeholders can be linked to UIT and SKM.

At last, in addition to the technical and methodological limitations mentioned in this study, developers also mentioned barriers related to the access of data and ethical preoccupation related to data privacy. This aligns with other researches in Latin America [37], [63] that also identified bureaucratic barriers such as the approval and certification of research, as well as difficulties in accessing data quickly and easily. These factors can hinder the use of the tool in a more widespread and efficient manner, limiting its potential for effective intervention in the teaching and learning process. The significance of aligning learning analytics (LA) development with ethical considerations, particularly in relation to data privacy, has been extensively highlighted in the literature [45], [56], [72], [73], and [74]. Trust concerns of teaching staff are specially concentrated on third parties involved with LA and how they handle privacy and ethicsrelated issues [75]. In the context of Brazil, this involves taking into account the General Data Protection Law, which the interviewed developers cite as a guiding framework in their investigations, development, and enhancement of LA tools.

#### **VI. CONCLUSION**

The present study investigated the adoption process of Brazilian learning analytics tools from the perspective of their developers. For that, six developers of LA tools were interviewed about the functionalities of their LA tools, potential applications, and limitations in the context of Brazil. The analysis were was conducted according to two main dimensions, which are: 1) The characteristics of the LA tools and 2) The potentialities and limitations of their adoption. As stated in during the paper, LA tools developed in Brazil and selected for this study are used in both face-to-face and distance learning, and their functionalities include predicting students at-risk of dropping out, providing automated and personalized correction and feedback, and supporting didactic planning of courses or disciplines. The intended users of these tools vary depending on the specific tool, and may include students, professors, or management groups.

Developers underscore the significant potential of these tools for interventions and driving transformative change in the applied context, ultimately enhancing the teaching and learning processes. In elucidating the tools' potential, developers specifically emphasized the following aspects: a) seamless integration with academic systems; b) enhanced monitoring of students; c) facilitation of teaching actions; d) improved visualization and interpretation of analyses through dashboards; e) streamlined intervention planning with the capacity to effect real change; and f) reduction in dropout rates and improvement in academic performance.

Nevertheless, developers also recognize various limitations impeding the broader adoption of learning analytics tools, including: a) the sociodemographic inequality inherent in the Brazilian context and the insufficient reach of the tools; b) inadequate support for technological innovation research, marked by yearly reductions in funding; c) restricted application of the tool to specific knowledge domains; d) insufficient feedback to students, as most analyses remain accessible only to teachers and administrators; e) a lack of awareness among the population regarding the importance of learning analytics and its potential impact on the teaching and learning process.

The paper has also examined how the features of the tools, their potentials, and limitations align with the current state of the field both in Latin America and globally. This is particularly noticeable as the present study presents the perspectives of the developers while the existing literature normally focuses on the opinions other stakeholders such as the teachers, the students and the managers. To the best of our knowledge, this is the first work tackling this specific group.

While this research provides valuable perspectives, there are also some limitations and areas for improvement. As the research was qualitative it is not possible to extrapolate the results found here to the whole country and region. Thus, future work should focus on quantitative approaches with a larger sample of developers and creators of other LA tools. It would also be interesting to follow up the evolution of the studied tools and contrasting the findings of the present work with data gathered with the real users of these tools, and how their interventions helped to improve learning outcomes. It would also be interesting to complement the sentiment analysis with contextual and cultural aspects of sentiment. Finally, the research could also be expanded to other countries in Latin America.

The insights presented in this paper can serve as a reference to strengthen the Learning Analytics field in Brazil and across Latin America, offering a more direct approach to addressing the challenges and issues highlighted by the developers of these tools in the region.

For future work, there is an opportunity to complement this work with an in-depth analysis of the strengths and weaknesses of various tools in educational contexts, utilizing relevant data. This approach promises to improve our understanding of the effectiveness of learning analytics tools and guide their more effective implementation in educational environments.

#### REFERENCES

- [1] A. C. O. Silva, S. D. A. Sousa, and J. B. F. D. Menezes, "O ensino remoto na percepção discente: Desafios e benefícios," *Dialogia*, no. 36, pp. 298–315, Sep./Dec. 2020. [Online]. Available: https://periodicos. uninove.br/dialogia/article/view/18383, doi: 10.5585/dialogia.n36.18383.
- [2] A. R. Masalimova, M. A. Khvatova, L. S. Chikileva, E. P. Zvyagintseva, V. V. Stepanova, and M. V. Melnik, "Distance learning in higher education during COVID-19," *Frontiers Educ.*, vol. 7, p. 120, Mar. 2022.
- [3] D. Ifenthaler and C. Widanapathirana, "Development and validation of a learning analytics framework: Two case studies using support vector machines," *Technol., Knowl. Learn.*, vol. 19, nos. 1–2, pp. 221–240, Jul. 2014.
- [4] D. Ifenthaler, D. Gibson, D. Prasse, A. Shimada, and M. Yamada, "Putting learning back into learning analytics: Actions for policy makers, researchers, and practitioners," *Educ. Technol. Res. Develop.*, vol. 69, no. 4, pp. 2131–2150, Aug. 2021.
- [5] G. Lynch, "13<sup>th</sup> international learning analytics and knowledge conference (LAK23)," ACM SIGWEB Newslett., vol. 2023, pp. 1–3, Oct. 2023.
- [6] S. Macneill, J. Mutton, W. Kraan, D. Sherlock, D. Griffiths, A. Cooper, and S. P. Iec, "CETIS analytics series," Tech. Rep., 2012.
- [7] L. Ali, M. Hatala, D. Gasević, and J. Jovanovic, "A qualitative evaluation of evolution of a learning analytics tool," *Comput. Educ.*, vol. 58, no. 1, pp. 470–489, Jan. 2012.
- [8] J. B. C. Nunes, H. Schneider, and G. Carvalho, A Analítica Da Aprendizagem: Contribuições Tecnologias Digitais Para a Educação Superior. Brazil: Editora Criação, 2016, pp. 45–56.
- [9] Y.-S. Tsai and D. Gasevic, "Learning analytics in higher education— Challenges and policies: A review of eight learning analytics policies," in *Proc. 7th Int. Learn. Analytics Knowl. Conf.*, 2017, pp. 233–242.
- [10] Y.-S. Tsai, D. Rates, P. M. Moreno-Marcos, P. J. Muñoz-Merino, I. Jivet, M. Scheffel, H. Drachsler, C. Delgado Kloos, and D. Gsevic, "Learning analytics in European higher education—Trends and barriers," *Comput. Educ.*, vol. 155, Oct. 2020, Art. no. 103933.
- [11] C. Colvin, T. Rodgers, A. Wade, S. Dawson, D. Gasevic, S. B. Shum, K. Nelson, S. Alexander, L. Lockyer, G. Kennedy, L. Corrin, and J. Fisher, "Student retention and learning analytics: A snapshot of Australian practices and a framework for advancement," Austral. Government Office Learn. Teach., Canberra, ACT, Australia, 2016. [Online]. Available: http://www.olt.gov.au/project-student-retention-andlearning-analytics-snapshot-currentaustralian-practices-and-framework
- [12] I. Hilliger, M. Ortiz-Rojas, P. Pesántez-Cabrera, E. Scheihing, Y.-S. Tsai, P. J. Muñoz-Merino, T. Broos, A. Whitelock-Wainwright, and M. Pérez-Sanagustín, "Identifying needs for learning analytics adoption in Latin American universities: A mixed-methods approach," *Internet Higher Educ.*, vol. 45, Apr. 2020, Art. no. 100726.
- VOLUME 12, 2024

- [13] T. P. Falcão, R. F. Mello, R. L. Rodrigues, J. R. B. Diniz, Y.-S. Tsai, and D. Gasevic, "Perceptions and expectations about learning analytics from a Brazilian higher education institution," in *Proc. 10th Int. Conf. Learn. Analytics Knowl.*, 2020, pp. 240–249.
- [14] L. Tomczyk, V. F. Martins, M. A. Eliseo, I. F. Silveira, C. D. L. H. Amato, and L. Stosic, "ICT and education in Brazil–NGO, local government administration, business and higher education expert perspective," *World J. Educ. Technol., Current Issues*, vol. 12, no. 4, pp. 401–424, Oct. 2020.
- [15] C. Cechinel, X. Ochoa, H. Lemos dos Santos, J. B. C. Nunes, V. Rodés, and E. Marques Queiroga, "Mapping learning analytics initiatives in Latin America," *Brit. J. Educ. Technol.*, vol. 51, no. 4, pp. 892–914, Jul. 2020.
- [16] T. F. Scheneider, C. Cechinel, R. F. L. D. Mello, E. L. S. X. Freitas, T. P. D. R. Falcão, and V. F. C. Ramos, "Análise do nível de maturidade na adoção de learning analytics em instituições de ensino superior do sul e sudeste brasileiro," *Revista Brasileira de Informática na Educação*, vol. 31, pp. 1005–1030, Dec. 2023.
- [17] A. van Leeuwen, "Teachers' perceptions of the usability of learning analytics reports in a flipped university course: When and how does information become actionable knowledge?" *Educ. Technol. Res. Develop.*, vol. 67, no. 5, pp. 1043–1064, Oct. 2019.
- [18] B. Rienties and L. Toetenel, "The impact of learning design on student behaviour, satisfaction and performance: A cross-institutional comparison across 151 modules," *Comput. Hum. Behav.*, vol. 60, pp. 333–341, Jul. 2016.
- [19] D. Gasevic, S. Joksimovic, B. R. Eagan, and D. W. Shaffer, "SENS: Network analytics to combine social and cognitive perspectives of collaborative learning," *Comput. Hum. Behav.*, vol. 92, pp. 562–577, Mar. 2019.
- [20] D. T. Tempelaar, B. Rienties, and B. Giesbers, "In search for the most informative data for feedback generation: Learning analytics in a data-rich context," *Comput. Hum. Behav.*, vol. 47, pp. 157–167, Jun. 2015.
- [21] J. Wong, M. Baars, B. B. de Koning, T. van der Zee, D. Davis, M. Khalil, G.-J. Houben, and F. Paas, "Educational theories and learning analytics: From data to knowledge: The whole is greater than the sum of its parts," in *Utilizing Learning Analytics to Support Study Success*. Cham, Switzerland: Springer, 2019, pp. 3–25.
- [22] A. S. Alzahrani, Y.-S. Tsai, S. Iqbal, P. M. M. Marcos, M. Scheffel, H. Drachsler, C. D. Kloos, N. Aljohani, and D. Gasevic, "Untangling connections between challenges in the adoption of learning analytics in higher education," *Educ. Inf. Technol.*, vol. 28, no. 4, pp. 4563–4595, Apr. 2023.
- [23] D. Gasevic, Y.-S. Tsai, and H. Drachsler, "Learning analytics in higher education—Stakeholders, strategy and scale," *Internet Higher Educ.*, vol. 52, Jan. 2022, Art. no. 100833.
- [24] H. L. Dos Santos, C. Cechinel, J. B. C. Nunes, and X. Ochoa, "An initial review of learning analytics in Latin America," in *Proc. 12th Latin Amer. Conf. Learn. Technol. (LACLO)*, 2017, pp. 1–9.
- [25] C. Cobo and C. Aguerrebere, "Building capacity for learning analytics in Latin America," *Include Us All! Directions for Adoption learning Analytics Global South*, vol. 58. Quezon, Philippines: Foundation for Information Technology Education and Development, 2018.
- [26] X. Ochoa, "Learning analytics in Latin America present an opportunity not to be missed," *Nature Hum. Behaviour*, vol. 3, no. 1, pp. 6–7, Nov. 2018.
- [27] J. Maldonado, I. Hilliger, T. De Laet, M. Millecamp, K. Verbert, X. Ochoa, and M. Pérez-Sanagustín, "The lala project: Building capacity to use learning analytics to improve higher education in Latin America," in *Proc.* 8th Int. Learn. Anal. Knowl. Conf., May 2018, pp. 630–637.
- [28] J. Blanc-Talon and K. Szczypiorski, "Preface," in Proc. 5th Int. Conf. Frontiers Signal Process. (ICFSP), Sep. 2019, pp. 1–7.
- [29] T. P. Falcão, R. F. Mello, and R. L. Rodrigues, "Applications of learning analytics in Latin America," *Brit. J. Educ. Technol.*, vol. 51, no. 4, pp. 871–874, 2020.
- [30] Y.-S. Tsai, P. M. Moreno-Marcos, K. Tammets, K. Kollom, and D. Gasevic, "SHEILA policy framework: Informing institutional strategies and policy processes of learning analytics," in *Proc. 8th Int. Conf. Learn. Analytics Knowl.*, Mar. 2018, pp. 320–329.
- [31] V. Kovanovic, C. Mazziotti, and J. Lodge, "Learning analytics for primary and secondary schools," J. Learn. Anal., vol. 8, no. 2, pp. 1–5, Sep. 2021.
- [32] S. Fan, L. Chen, M. Nair, S. Garg, S. Yeom, G. Kregor, Y. Yang, and Y. Wang, "Revealing impact factors on student engagement: Learning analytics adoption in online and blended courses in higher education," *Educ. Sci.*, vol. 11, no. 10, p. 608, Oct. 2021.

- [33] R. Kaliisa, A. Kluge, and A. I. Mørch, "Overcoming challenges to the adoption of learning analytics at the practitioner level: A critical analysis of 18 learning analytics frameworks," *Scandin. J. Educ. Res.*, vol. 66, no. 3, pp. 367–381, Apr. 2022.
- [34] E. Freitas, F. Fonseca, V. Garcia, T. P. Falcao, E. Marques, D. Gašević, and R. F. Mello, "MMALA: Developing and evaluating a maturity model for adopting learning analytics," *J. Learn. Anal.*, vol. 11, no. 1, pp. 67–86, 2024, doi: 10.18608/jla.2024.8099.
- [35] T. F. Scheneider, C. Cechinel, R. F. L. D. Mello, E. L. S. X. Freitas, T. P. D. R. Falcão, and V. F. C. Ramos, "Análise do nível de maturidade na adoção de learning analytics em instituições de ensino superior brasileiras," in *Proc. Anais do I Workshop de Aplicações Práticas de Learning Analytics em Instituições de Ensino no Brazil (WAPLA)*, Nov. 2022, pp. 11–20.
- [36] E. G. Guba and Y. S. Lincoln, "Competing paradigms in qualitative research," in *Handbook of Qualitative Research*, vol. 2. Newbury Park, CA, USA: Sage, 1994, p. 105.
- [37] R. H. Sampieri, C. F. Collado, and P. B. Lucio, *Metodología de la Investigación*, vol. 4. New York, NY, USA: McGraw-Hill, 2018.
- [38] G. R. Gibbs, Analyzing Qualitative Data (Qualitative Research Kit). London, U.K.: Sage, 2007, doi: 10.4135/9781849208574.
- [39] L. Bardin, Análise de Conteúdo, 1st ed. São, Paulo: Edições, 2015.
- [40] S. Elo, M. Kääriäinen, O. Kanste, T. Pölkki, K. Utriainen, and H. Kyngäs, "Qualitative content analysis: A focus on trustworthiness," *SAGE Open*, vol. 4, no. 1, Jan. 2014, Art. no. 215824401452263, doi: 10.1177/2158244014522633.
- [41] J. N. Lester, Y. Cho, and C. R. Lochmiller, "Learning to do qualitative data analysis: A starting point," *Hum. Resource Develop. Rev.*, vol. 19, no. 1, pp. 94–106, Mar. 2020, doi: 10.1177/1534484320903890.
- [42] M. Gläser-Zikuda, G. Hagenauer, and M. Stephan, "The potential of qualitative content analysis for empirical educational research," *Forum Qualitative Sozialforschung/Forum, Qualitative Social Res.*, vol. 21, no. 1, p. 20, Jan. 2020. [Online]. Available: https://www.qualitativeresearch.net/index.php/fqs/article/view/3443
- [43] C. Cechinel, D. Dermeval, E. Oliveira, I. Gasparini, R. Araujo, and R. F. Mello, "Editorial do wapla 2022," in Anais do I Workshop de Aplicações Práticas de Learning Analytics em Instituições de Ensino no Brasil. Porto Alegre, RS, Brasil: SBC, 2022, pp. 1–11. [Online]. Available: https://sol.sbc.org.br/index.php/wapla/article/view/22516
- [44] B. D. G. E. Estatistica, "Pesquisa nacional por amostra de domicílios contínua," Instituto Brasileiro de Geografia e Estadistica (IBGE), Brazil, Tech. Rep., 2020. [Online]. Available: https://webapps.ilo.org/survey Lib/index.php/catalog/6412
- [45] M. J. Rodríguez-Triana, A. Martínez-Monés, and S. Villagrá-Sobrino, "Learning analytics in small-scale teacher-led innovations: Ethical and data privacy issues," *J. Learn. Anal.*, vol. 3, no. 1, pp. 43–65, Apr. 2016.
- [46] Y. S. O. Sousa, "O uso do software iramuteq: Fundamentos de lexicometria para pesquisas qualitativas," *Estudos e Pesquisas em Psicologia*, vol. 21, no. 4, pp. 1541–1560, Dec. 2021.
- [47] A. N. de Vasconcelos, L. A. Freires, G. D. L. Loureto, G. Fortes, J. C. A. da Costa, L. F. F. Torres, I. I. Bittencourt, T. D. Cordeiro, and S. Isotani, "Advancing school dropout early warning systems: The IAFREE relational model for identifying at-risk students," *Frontiers Psychol.*, vol. 14, Jul. 2023, Art. no. 1189283.
- [48] W. L. Cambruzzi, "GVwise: Uma aplicação de learning analytics para a redução da evasão na educação a distância," Dissertação (mestrado)— Programa de Pós-graduação em Computação Aplicada, Universidade do Vale do Rio dos Sinos, São Leopoldo, Brazil, 2014.
- [49] S. J. Rigo, W. Cambruzzi, J. L. V. Barbosa, and S. C. Cazella, "Aplicações de mineração de dados educacionais e learning analytics com foco na evasão escolar: Oportunidades e desafios," *Revista Brasileira de Informática na Educação*, vol. 22, no. 1, p. 132, May 2014.
- [50] L. Galvão, D. Fernandes, and B. Gadelha, "Juiz online como ferramenta de apoio a uma metodologia de ensino híbrido em programação," in *Proc. Anais do 27th Simpósio Brasileiro de Informática na Educação (SBIE)*, Nov. 2016, p. 140.
- [51] H. L. Paraiba, J. B. Escorel, P. A. Vieira, R. Britto, W. L. Soares, I. D. Nunes, and U. Schiel, "ATID—Authoring tool for instructional design," in *Proc. 13th FISL-Forum Internacional de Softw. Livre. Porto Alegre*, 2012, pp. 383–389.
- [52] G. A. N. N. Diniz, E. E. D. S. Alencar, and I. D. Nunes, "Monitores automáticos—Uso de learning analytics para acompanhamento de atividades," in Proc. Anais do I Workshop de Aplicações Práticas de Learn. Analytics em Instituições de Ensino no Brazil (WAPLA), Nov. 2022, pp. 79–87.

- [53] C. Cechinel, T. Primo, M. S. Stein, V. C. Ramos, M. Machado, C. Barrozo, E. Queiroga, J. Rocha, H. Lemos, and D. Moro, "Solução para a predição de risco académico através de aprendizado de máquina," in *Proc. WRNP*, 2022, pp. 26–27.
- [54] R. Ferreira Mello, G. Fiorentino, H. Oliveira, P. Miranda, M. Rakovic, and D. Gasevic, "Towards automated content analysis of rhetorical structure of written essays using sequential content-independent features in Portuguese," in *Proc. 12th Int. Learn. Analytics Knowl. Conf.*, Mar. 2022, pp. 404–414.
- [55] B. A. B. Rosa and R. Ferreira Mello, "Análise automatizada de coesão em redações do ensino fundamental por meio de técnicas de processamento de linguagem natural," in *Proc. Anais Estendidos do 11th Congresso Brasileiro de Informática na Educação (CBIE)*, Nov. 2022, pp. 144–149.
- [56] J. B. F. D. Menezes, A. C. A. D. Almeida, M. V. D. A. Lima, and T. M. Sousa, "Ética e learning analytics: Uma combinação necessária para o rigor científico," *Práxis Educativa*, vol. 18, pp. 1–11, 2023, Art. no. e21370.
- [57] F. Barbieri, L. Espinosa Anke, and J. Camacho-Collados, "XLM-T: Multilingual language models in Twitter for sentiment analysis and beyond," in *Proc. 13th Lang. Resour. Eval. Conf.*, Marseille, France:, Jun. 2022, pp. 258–266. [Online]. Available: https://aclanthology.org/2022.lrec-1.27
- [58] A. Sales, L. Balby, and A. Cajueiro, "Exploiting academic records for predicting student drop out: A case study in Brazilian higher education," *J. Inf. Data Manag.*, vol. 7, no. 2, p. 166, 2016.
- [59] I. Lykourentzou, I. Giannoukos, V. Nikolopoulos, G. Mpardis, and V. Loumos, "Dropout prediction in e-learning courses through the combination of machine learning techniques," *Comput. Educ.*, vol. 53, no. 3, pp. 950–965, Nov. 2009.
- [60] S. M. Jayaprakash, E. W. Moody, E. J. M. Lauría, J. R. Regan, and J. D. Baron, "Early alert of academically at-risk students: An open source analytics initiative," *J. Learn. Analytics*, vol. 1, no. 1, pp. 6–47, May 2014.
- [61] E. M. Queiroga, J. L. Lopes, K. Kappel, M. Aguiar, R. M. Araujo, R. Munoz, R. Villarroel, and C. Cechinel, "A learning analytics approach to identify students at risk of dropout: A case study with a technical distance education course," *Appl. Sci.*, vol. 10, no. 11, p. 3998, Jun. 2020. [Online]. Available: https://www.mdpi.com/2076-3417/10/11/3998
- [62] E. M. Queiroga, C. R. Enríquez, C. Cechinel, A. P. Casas, V. R. Paragarino, L. R. Bencke, and V. F. C. Ramos, "Using virtual learning environment data for the development of institutional educational policies," *Appl. Sci.*, vol. 11, no. 15, p. 6811, Jul. 2021.
- [63] L. A. Buschetto Macarini, C. Cechinel, M. F. Batista Machado, V. Faria Culmant Ramos, and R. Munoz, "Predicting students success in blended learning—Evaluating different interactions inside learning management systems," *Appl. Sci.*, vol. 9, no. 24, p. 5523, Dec. 2019.
- [64] A. Pardo, J. Jovanovic, S. Dawson, D. Gasevic, and N. Mirriahi, "Using learning analytics to scale the provision of personalised feedback," *Brit. J. Educ. Technol.*, vol. 50, no. 1, pp. 128–138, Jan. 2019.
- [65] H. E. Schaffer, K. R. Young, E. W. Ligon, and D. D. Chapman, "Automating individualized formative feedback in large classes based on a directed concept graph," *Frontiers Psychol.*, vol. 8, p. 260, Feb. 2017.
- [66] M. Ez-zaouia, A. Tabard, and E. Lavoué, "Emodash: A dashboard supporting retrospective awareness of emotions in online learning," *Int. J. Hum.-Comput. Stud.*, vol. 139, Jul. 2020, Art. no. 102411.
- [67] R. Kaliisa, B. Rienties, A. I. Mørch, and A. Kluge, "Social learning analytics in computer-supported collaborative learning environments: A systematic review of empirical studies," *Comput. Educ. Open*, vol. 3, Dec. 2022, Art. no. 100073.
- [68] Sociedade Brasileira para o Progresso da Ciência (SBPC), "Projeto para um Brasil Novo," 2014. [Online]. Available: https://portal.sbpcnet. org.br/projeto-brasil-novo
- [69] C. Klein, J. Lester, H. Rangwala, and A. Johri, "Technological barriers and incentives to learning analytics adoption in higher education: Insights from users," J. Comput. Higher Educ., vol. 31, no. 3, pp. 604–625, Dec. 2019.
- [70] P. A. Kirschner and P. De Bruyckere, "The myths of the digital native and the multitasker," *Teaching Teacher Educ.*, vol. 67, pp. 135–142, Oct. 2017. [Online]. Available: https://www.sciencedirect.com/science/article/pii/ S0742051X16306692
- [71] L. Márquez, V. Henríquez, H. Chevreux, E. Scheihing, and J. Guerra, "Adoption of learning analytics in higher education institutions: A systematic literature review," *Brit. J. Educ. Technol.*, vol. 55, no. 2, pp. 439–459, Mar. 2024. [Online]. Available: https://berajournals.onlinelibrary.wiley.com/doi/abs/10.1111/bjet.13385
- [72] M. Khalil, P. Prinsloo, and S. Slade, "Fairness, trust, transparency, equity, and responsibility in learning analytics," *J. Learn. Anal.*, vol. 10, no. 1, pp. 1–7, Mar. 2023.

- [73] E. Gedrimiene, I. Celik, K. Mäkitalo, and H. Muukkonen, "Transparency and trustworthiness in user intentions to follow career recommendations from a learning analytics tool," *J. Learn. Analytics*, vol. 10, no. 1, pp. 54–70, Mar. 2023.
- [74] I. Rets, C. Herodotou, and A. Gillespie, "Six practical recommendations enabling ethical use of predictive learning analytics in distance education," *J. Learn. Anal.*, vol. 10, no. 1, pp. 149–167, Feb. 2023.
- [75] A. S. Alzahrani, Y.-S. Tsai, N. Aljohani, E. Whitelock-wainwright, and D. Gasevic, "Do teaching staff trust stakeholders and tools in learning analytics? A mixed methods study," *Educ. Technol. Res. Develop.*, vol. 71, no. 4, pp. 1471–1501, Aug. 2023.



JONES BARONI FERREIRA DE MENEZES received the master's degree in physiological sciences from the State University of Ceará and the Ph.D. degree in education. He is currently an Associate Professor I in the biological sciences course with the Faculty of Education in Crateús and a Teacher Trainer and an Extension Coordinator in the distance learning biological sciences course. His current research interests include science/biology education, educational

technologies, and distance education in the field of education. As a Permanent Faculty Member of the Professional Master's in Biology Teaching (PROFBIO), he brings expertise in Research Ethics Committee Coordination and Membership.



**EMANUEL MARQUES QUEIROGA** received the Graduate degree in systems analysis and development from the Catholic University of Pelotas (UCPel) and the master's and Ph.D. degrees in computer science from the Federal University of Pelotas (UFPel). He is currently a Brazilian Researcher. He is also an Analyst with the Federal Institute of Education, Science, and Technology of Southern Rio Grande (IFSul). Simultaneously, he develops projects in learning analytics and

educational data mining in Latin America, focusing on the modeling and application of prediction methods in educational data. His research interests include data science, data mining, big data, machine learning, and learning analytics.



**ROBERTO MUÑOZ SOTO** received the master's degree in computer engineering, in engineering science, and in higher education and the Ph.D. degree in computer engineering. He is currently a Full Professor and the Head of the School of Informatics Engineering, Universidad de Valparaíso, Chile. His research interests include computers and education, multimodal learning analytics, human–computer interaction, and health informatics.



VIRGINIA RODÉS PARAGARINO received the Ph.D. degree in equity and innovation and in education from University de Santiago de Compostela. She is currently a Postdoctoral Fellow with the Educational Technology Unit, Institute for the Future of Education (Tecnológico de Monterrey). Her research interests include the study, development, and adoption of educational technologies in higher education, with an emphasis on open education. She is also a

member of the National System of Researchers of Uruguay in Social Sciences-Communication and Media. She holds the UNESCO Chair of Open Education (Uruguay), is a member of the LATAM Regional Node of Open Education Global and the Open Education for a Better World Advisory Board, and is a fellow of the Global OER Graduate Network (GO-GN).



VINÍCIUS FARIA CULMANT RAMOS received the bachelor's degree in computer science from UFRJ, the master's degree in systems and computing engineering from COPPE/UFRJ, and the joint Ph.D. degree in systems and computing engineering with a co-tutelage agreement between COPPE/UFRJ and the Eindhoven University of Technology, The Netherlands. He is currently an Associate Professor with the Department of Knowledge Engineering, Federal University of

Santa Catarina (UFSC). He is also a Professor with the Postgraduate Programs in Information and Communication Technologies (PPGTIC/UFSC) and in Engineering and Knowledge Management (PPGEGC). His research interests include intelligent educational systems, learning analytics, computer education, data science applied to education and business, and social network data analysis.



**LYDIA MENESES DE CARVALHO** is currently pursuing the bachelor's degree in information and communication technologies with the Federal University of Santa Catarina. She is also a Scientific and Technological Initiation Scholarship Holder from the National Council for Scientific and Technological Development (CNPq). Her research interests include educational data visualizations and learning analytics.



AUGUSTO DE CARVALHO FERNANDES DE SOUZA is currently pursuing the bachelor's degree in information and communication technologies with the Federal University of Santa Catarina. His research interests include educational data visualizations and learning analytics. He has experience in software development with a specialization in front-end using JavaScript, CSS, and HTML. He is also a Scientific and Technological Initiation Scholarship Holder from

the National Council for Scientific and Technological Development (CNPq).



**CRISTIAN CECHINEL** received the Ph.D. degree in information and knowledge engineering from the University of Alcalá, Spain. He is currently an Associate Professor with the Sciences, Technologies and Health Center, Federal University of Santa Catarina, Brazil. His research interests include the development and analysis of digital learning technologies, learning analytics, and distance learning. He is also a CNPq Fellow of Research Productivity in Technological Develop-

ment and Innovative Extension (DT2).

...